Author's response to reviews

Title: A Preliminary study of p53 and c-erbB-2 expression in gallbladder cancer in Indian patients

Authors:

Amita Chaube (amitachaube@yahoo.com)
Mallika Tewari (mallika_vns@usa.net.in)
Usha Singh (sonarita001@yahoo.co.in)
Rajendra S Garbyal (rajgarbyal@yahoo.com)
Hari S Shukla (harishukla@usa.net)

Version: 5 Date: 10 March 2006

Author’s response to reviews:

1. We have carried out formatting changes in according with the requirement of the Journal. The formatting changes have been incorporated into the fresh manuscript. The manuscript has been revised conforming to your instructions.
2. We are returning you the revised manuscript on 8 March, well before the deadline of 13th March 2006.
3. The authorship is aware that a processing charge of GBP 375 will have to be paid of the article is to be accepted.
4. With reference to referee 1: this work is a product of PhD thesis being conducted in the Institute of Medical Sciences, Banaras Hindu University. All the work conforms to the universities sets standards including consideration of ethics and consent. There was no animal experimentation involved.
5. With reference to reviewer Nick J Agnantis :- we have changed the title of our paper:-

This is preliminary study that we have carried out therefore more cases cannot be taken at this stage.

In the discussion section relevance of cytoplasmic c-erbB-2 staining have been included.